

W&M ScholarWorks

## **VIMS Articles**

Virginia Institute of Marine Science

2017

# Preliminary estimate of contribution of Arctic nitrogen fixation to the global nitrogen budget

RE Sipler Virginia Institute of Marine Science

Donglai Gong Virginia Institute of Marine Science

SE Baer Virginia Institute of Marine Science

MP Sanderson Virginia Institute of Marine Science

QN Roberts Virginia Institute of Marine Science

See next page for additional authors

Follow this and additional works at: https://scholarworks.wm.edu/vimsarticles

🔮 Part of the Marine Biology Commons, and the Oceanography Commons

### **Recommended Citation**

Sipler, RE; Gong, Donglai; Baer, SE; Sanderson, MP; Roberts, QN; Mulholland, M; and Bronk, DA, "Preliminary estimate of contribution of Arctic nitrogen fixation to the global nitrogen budget" (2017). *VIMS Articles*. 1436. https://scholarworks.wm.edu/vimsarticles/1436

This Article is brought to you for free and open access by the Virginia Institute of Marine Science at W&M ScholarWorks. It has been accepted for inclusion in VIMS Articles by an authorized administrator of W&M ScholarWorks. For more information, please contact scholarworks@wm.edu.

## Authors

RE Sipler, Donglai Gong, SE Baer, MP Sanderson, QN Roberts, M Mulholland, and DA Bronk





*Limnology and Oceanography Letters* 2, 2017, 159–166 © 2017 The Author. Limnology and Oceanography Letters published by Wiley Periodicals, Inc. on behalf of Association for the Sciences of Limnology and Oceanography doi: 10.1002/lol2.10046

## LETTER

## Preliminary estimates of the contribution of Arctic nitrogen fixation to the global nitrogen budget

Rachel E. Sipler <sup>(D)</sup>,<sup>1</sup>\* Donglai Gong,<sup>1</sup> Steven E. Baer <sup>(D)</sup>,<sup>1,2</sup> Marta P. Sanderson,<sup>1</sup> Quinn N. Roberts,<sup>1</sup> Margaret R. Mulholland,<sup>3</sup> Deborah A. Bronk<sup>1</sup>

<sup>1</sup>The Virginia Institute of Marine Science, College of William & Mary, Gloucester Point, Virginia; <sup>2</sup>Bigelow Laboratory for Ocean Sciences, East Boothbay, Maine; <sup>3</sup>Department of Ocean, Earth and Atmospheric Sciences, Old Dominion University, Norfolk, Virginia

### Scientific Significance Statement

Nitrogen fixation is an important source of "new" nitrogen to both freshwater and marine ecosystems where it supports biological production and balances aquatic nitrogen budgets. Although freshwater nitrogen fixation is relatively common at high latitudes, water column marine nitrogen fixation has historically been considered to be a mostly warm water, oligotrophic process. Here, we provide evidence to show that marine nitrogen fixation could be an important source of nitrogen to the seasonally nitrogen-limited Arctic Ocean, which could have far reaching implications for primary productivity and biogeochemical budgets in this rapidly changing ecosystem.

## Abstract

Dinitrogen  $(N_2)$  fixation is the source of all biologically available nitrogen on earth, and its presence or absence impacts net primary production and global biogeochemical cycles. Here, we report rates of 3.5-17.2 nmol N L<sup>-1</sup> d<sup>-1</sup> in the ice-free coastal Alaskan Arctic to show that N<sub>2</sub> fixation in the Arctic Ocean may be an important source of nitrogen to a seasonally nitrogen-limited system. If widespread in surface waters over ice-free shelves throughout the Arctic, N<sub>2</sub> fixation could contribute up to 3.5 Tg N yr<sup>-1</sup> to the Arctic nitrogen budget. At these rates, N<sub>2</sub> fixation occurring in ice-free summer waters would offset up to 27.1% of the Arctic denitrification deficit and contribute an additional 2.7% to N<sub>2</sub> fixation globally, making it an important consideration in the current debate of whether nitrogen in the global ocean is in steady state. Additional investigations of high-latitude marine diazotrophic physiology are required to refine these  $N_2$  fixation estimates.

Net primary productivity in the Arctic Ocean is projected to increase in response to higher temperatures and consequent ice melt (Arrigo et al. 2012). Any increase in productivity, however, will require sufficient nitrogen to support it (Tremblay and Gagnon 2009). While several studies have speculated on changes to the Arctic nitrogen cycle and resultant effects on the primary producers that form the base of the food chain (Tremblay et al. 2008; Popova et al. 2012), the role of dinitrogen (N<sub>2</sub>) fixation in the Arctic Ocean has received comparatively little attention.

Historically, marine N2 fixation has been considered a warm-water process that occurs mainly in subtropical oligotrophic gyres and tropical seas, where it provides a new source of nitrogen to the system (Sohm et al. 2011). Recent work has shown, however, that N<sub>2</sub> fixers (Moisander et al.

<sup>\*</sup>Correspondence: rachelsipler@gmail.com

Author Contribution Statement: DAB and RES designed the experiment. RES and MPS performed the experiments. RES, DAB, MPS, SEB, QNR, and MRM analyzed the data. DG determined the area and duration of ice-free regions. RES and DG calculated the potential contributions of N<sub>2</sub> fixation to local and global nitrogen budgets. RES and DAB wrote the paper with input from all authors.

Data Availability Statement: Data are available in the Biological & Chemical Oceanography Data Management Office (BCO-DMO) repository at http://www.bco-dmo.org/dataset/701789/data.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Station	Longitude (N)	Latitude (W)	Water column depth (m)	Sample collection depth (m)	Time started (local)	Temperature (°C)	Salinitv
Marine		()			(10 000)	(-)	
1	71°20′40″	156°41′25″	17.1	4	19:10	5.0	30.5
2	71°20′40″	156°41′25″	17.5	8	15:16	4.7	30.4
3	71°20′36″	156°39′30″	8.0	2	20:38	4.6	30.4
4	71°20′40″	156°41′25″	17.5	8	23:57	4.5	30.4
8	71°19′02″	156°50′18″	ND	8	10:30	ND	30.4
Estuarine							
5	71°7′8″	156°54′24″	ND	Surface	15:53	6.7	12.0
6	71°7′45″	157°1′10″	ND	Surface	16:04	6.2	15.0
7	71°8′45″	157°3′45″	ND	Surface	16:18	6.0	20.1

Marine stations were located in the coastal Chukchi Sea (2.5 km northwest of Barrow, Alaska) and estuarine stations were in Walakpa Bay (20 km southwest of Barrow, Alaska). The local time of day when incubations was started is presented based on as 24-h clock.

2010; Blais et al. 2012; Díez et al. 2012; Fernández-Méndez et al. 2016) and  $N_2$  fixation (Blais et al. 2012; Mulholland et al. 2012) can occur at lower temperatures and higher latitudes than once thought and in areas where nitrogen is not limiting (Hamersley et al. 2011; Sohm et al. 2011) shifting the paradigm of where marine  $N_2$  fixation occurs.

The nifH gene encodes for the nitrogenase enzymes, which is responsible for N<sub>2</sub> fixation processes. Targeted nifH gene and broad metagenomic assessments have revealed the genetic capability for N2 fixation in a variety of polar microbial communities. For example, nifH or species known to possess it have been detected in sea ice (e.g., Bowman et al. 2014; Fernández-Méndez et al. 2016), polar lakes (e.g., Toetz 1961; Shtarkman et al. 2013), the Arctic Ocean (e.g., Blais et al. 2012; Díez et al. 2012; Fernández-Méndez et al. 2016), glaciers (e.g., Telling et al. 2011), and polar soils (e.g., Solheim et al. 1996; Dickson 2000). These investigations provide compelling evidence that diverse polar microbial communities have the genes required to fix nitrogen. Fewer studies provide actual rates of marine N2 fixation that can be used to estimate the amount of nitrogen contributed to global nitrogen budgets (e.g., Knowles and Wishart 1977; Haines et al. 1981; Gihring et al. 2010; Blais et al. 2012).

The goal of this study was to determine the rate of  $N_2$  fixation in the surface mixed layer of the coastal Alaskan Arctic and to estimate how much nitrogen this process could contribute to local and global nitrogen budgets. These objectives were achieved using <sup>15</sup>N tracer techniques to measure rates of  $N_2$  fixation during the summer under ice-free conditions at marine and estuarine sites in the Chukchi Sea, near Barrow, Alaska. The data presented here, and reported in Blais et al. (2012), were extrapolated to provide rough estimates of the potential range of fixed nitrogen contributed by Arctic marine  $N_2$  fixation.

## Methods

#### N<sub>2</sub> fixation

Field samples were collected under ice-free conditions 15–20 August 2011, in the coastal waters of the Chukchi Sea (salinity of 30) and Walakpa Bay (salinity of 12–20) near Barrow, Alaska. Rates of N<sub>2</sub> fixation were measured using the <sup>15</sup>N bubble addition method (Montoya et al. 1996). Gastight 1L glass KIMAX<sup>TM</sup> media bottles (model # 611001000) capped with Wheaton<sup>TM</sup> black open-top caps with gray butyl septa (model # 240680) were used for all incubations. The glass media bottles were acid washed (10% HCl), rinsed four times with high purity water (18.2 M $\Omega$  cm<sup>-1</sup>) and combusted at 500°C for 4 h. The caps were submersed in a saltwater brine (approximately salinity of 60) for ~ 30 d, to condition them, and then acid washed (10% HCl) and rinsed with copious amounts of high purity water.

Seawater was collected using a low-pressure submersible electric pump (Johnson Pump model #16004) powered by a portable generator. Triplicate bottles were filled with seawater, capped and the ambient air bubbles were removed. The bottles were then amended with 1.2 mL (at a ratio of 1 mL of gas per 1L of seawater) of enriched (> 99%)<sup>15</sup>N<sub>2</sub> gas purchased from Cambridge Isotope Laboratories (lot #11-10077). Samples were incubated for 24 h in environmental chambers located at the Barrow Arctic Research Center. To mimic in situ conditions, the temperature was set at 4.5°C and light levels to 50  $\mu$ mol m<sup>-2</sup> s<sup>-1</sup> of light on a 24 h light cycle. No control incubation was done without added tracer gas. Sample locations, collection depths, and environmental conditions are presented in Table 1.

After 24 h, the incubations were terminated by filtering through 3.0  $\mu$ m silver filters or pre-combusted (450°C for 2 h) GF/F filters with a nominal pore size of 0.7  $\mu$ m. Filters were placed in sterile microcentrifuge tubes and stored

Sipler et al.

frozen at  $-20^{\circ}$ C. Prior to analysis, filters were thawed and dried at 40°C overnight. Based on cell counts by flow cytometry, approximately 61% of the bacterial cells were retained by the GF/F (described in more detail in Baer et al. 2017). Isotopic measurements for <sup>15</sup>N fixation rates were analyzed on a Europa GEO 20/20 mass spectrometer with an ANCA-SL autosampler. The atom % enrichments for all N2 fixation samples ranged from 0.3691 to 0.4463 and sample mass ranged from 9.25 µg N to 37.25 µg N. N<sub>2</sub> fixation rates were calculated using a mixing model (Montoya et al. 1996) and are reported as the mean ± the standard error for each site (n = 3) or grouped as marine (n = 15, five sites) or estuarine (n = 9, three sites) systems. The atom % enrichment of the particulate samples was measured against an atmospheric standard. They were not corrected for any variations in the natural abundance of the particulate material at the start of the incubation. This is a potential issue because particulate nitrogen has been shown to be enriched in <sup>15</sup>N in some regions of the coastal Arctic; in the summer of 2012,  $\delta^{15}$ N values of 6.66% and 7.78% were reported at coastal sites in the Chukchi to the west of the sample sites from this study (Yu et al. 2014). These  $\delta^{15}$ N values would not negate our observations.

We note that the bubble method has been shown to underestimate N<sub>2</sub> fixation based on gas solubility and may have biases against smaller symbiotic diazotrophs (Mohr et al. 2010; Großkopf et al. 2012). However, the overall magnitude of this underestimation is dependent upon the duration of the incubation and the temperature of the water in which the bubble is dissolved (Mohr et al. 2010). Culture experiments show that 75% of <sup>15</sup>N<sub>2</sub> gas had reached equilibrium after 24 h at considerably higher temperatures (28°C) using the bubble method, and result in only a 15% underestimation in N<sub>2</sub> fixation rates (Mohr et al. 2010). Unlike the bubble method, the direct injection method (Mohr et al. 2010) requires a 10% dilution of the microbial community with filtered <sup>15</sup>N<sub>2</sub> supersaturated seawater. Hypothesizing that fixation rates and diazotrophic abundance in the Arctic samples would be low, the bubble method was used to ensure that we did not dilute the community and therefore the rates of N<sub>2</sub> fixation. There have been no studies that report the bubble method overestimating N<sub>2</sub> fixation, therefore, the values reported here should be considered conservative estimates of N<sub>2</sub> fixation rates for the coastal Chukchi Sea.

We also note that contaminants ( ${}^{15}NH_4^+$ ,  ${}^{15}NO_3^-/NO_2^-$ ) have been measured in some  ${}^{15}N_2$  gas stocks (Dabundo et al. 2014). If contaminated  ${}^{15}N_2$  is used, measured N<sub>2</sub> fixation rates may be inflated due to the uptake of the contaminant and not fixation of the added  ${}^{15}N_2$  gas. Dabundo et al. (2014) compared the level of  ${}^{15}N$  gas contamination from three commonly used manufacturers of  ${}^{15}N_2$ . The  ${}^{15}N_2$  gas used in our study was purchased from Cambridge Isotope Laboratories, which had the lowest level of contamination of all sources tested. The highest levels of contamination reported for Cambridge Isotope Laboratories,  $^{15}\mathrm{N}_2$  gas were not substantial enough to negate the rates we observed in the Arctic and would have accounted for a negligible proportion ( $\leq 0.3\%$ ) of the rates observed.

#### Nutrients

Ammonium concentrations were analyzed using the colorimetric phenol-hypochlorite method (Koroleff 1983). Concentrations of nitrate, nitrite, and phosphate were measured using a Lachat QuikChem 8500 autoanalyzer (Parsons et al. 1984). Nutrient data are reported as the mean  $\pm$  the standard error of triplicate samples (Table 2).

#### Spatial and temporal calculations

A first order regional and pan-Arctic estimate of the amount of nitrogen that may be released via marine N<sub>2</sub> fixation was calculated using the total volume in the upper 50 m of the ice-free water column, integrated over the summer season (June-September), then multiplying that volume by the range of N<sub>2</sub> fixation rates observed in this and the Blais et al. (2012) studies. The extrapolation of the total nitrogen fixed under ice-free conditions between June and September is done for three domains: western Arctic shelves, pan Arctic continental shelves, and the entire Arctic Ocean. The western Arctic shelf is specified because the N<sub>2</sub> fixation rates reported here and in Blais et al. (2012), encompass the area between Baffin Bay on the northeastern boundary and the Chukchi Sea on the northwestern boundary (Fig. 1). For the purpose of this study, the western Arctic shelf was defined as bodies of water in the longitudinal range between 175°E and 45°W with water column depths shallower than 200 m and deeper than 1 m. Generally, this covers the Chukchi Sea, Beaufort shelves, Canadian Arctic Archipelago, the shelves surrounding Baffin Bay, and northern portion of the Labrador Sea. The Arctic continental shelves domain was defined as bodies of water that are not rivers or lakes north of latitude 64°N and with a water column depth shallower than 200 m and deeper than 1 m. The Arctic Ocean domain was defined as bodies of water that are not rivers or lakes north of latitude 64°N and with a water column depth of 1 m or greater. The 1 m shallow depth cut off was implemented to eliminate regions that are strongly waveinfluenced or very shallow tidal systems as measurements used in this study were not made in those types of environments.

For the volumetric calculations of seasonally ice-free waters, we used the 30 arcsec resolution IBCAO3 bathymetric data (Jakobsson et al. 2012) and Multisensor Analyzed Sea Ice Extent (MASIE) Northern Hemisphere data published by the National Snow and Ice Data Center (Fetterer et al. 2010). The MASIE sea ice cover data has a constant horizontal resolution of 4 km, with each grid covering an area of 16 km<sup>2</sup>. The climatological open water extent is calculated using MASIE data from 2006 to 2016.

	NU+	NO $_3^-$ ( $\mu$ mol N L $^{-1}$ )	PO <sub>4</sub> (μmol P L <sup>-1</sup> )	$N_2$ fixation rate (nmol N L <sup>-1</sup> d <sup>-1</sup> )		<b>5</b> 2
Station	$(\mu \text{mol N L}^{-1})$			Whole	> 3 µm	
Marine						
1	$1.55\pm0.07$	$0.45\pm0.00$	$0.55\pm0.01$	5.7 ± 1.7	$1.6\pm0.3$	34
2	$\textbf{0.59} \pm \textbf{0.00}$	$\textbf{0.32}\pm\textbf{0.00}$	$0.51\pm0.00$	$17.2\pm7.0$	$12.6\pm5.0$	84
3	$\textbf{0.47}\pm\textbf{0.02}$	$\textbf{0.33}\pm\textbf{0.00}$	$\textbf{0.47}\pm\textbf{0.00}$	$3.5\pm0.2$	$1.1 \pm 0.2$	31
4	$0.59\pm0.03$	$0.29\pm0.00$	$0.55\pm0.00$	$4.4\pm1.2$	$1.7\pm0.5$	40
8	$0.22\pm0.00$	$\textbf{0.30} \pm \textbf{0.00}$	$\textbf{0.39} \pm \textbf{0.00}$	$7.7 \pm 2.1$	$1.9\pm0.4$	25
Average	$\textbf{0.68} \pm \textbf{0.12}$	$\textbf{0.34} \pm \textbf{0.02}$	$\textbf{0.49} \pm \textbf{0.01}$	7.7±1.8	$\textbf{3.8} \pm \textbf{1.5}$	43
Estuarine						
5	$0.14\pm0.01$	$0.22\pm0.01$	$\textbf{0.17}\pm\textbf{0.00}$	$4.9\pm0.4$	$2.6\pm0.5$	52
6	$0.11\pm0.00$	$\textbf{0.59}\pm\textbf{0.00}$	$\textbf{0.17}\pm\textbf{0.01}$	$5.4\pm0.3$	$5.1\pm0.6$	96
7	$\textbf{0.10} \pm \textbf{0.00}$	$0.21\pm0.00$	$0.45\pm0.00$	$5.6\pm0.6$	$3.1\pm0.2$	57
Average	$\textbf{0.12} \pm \textbf{0.01}$	$\textbf{0.34} \pm \textbf{0.06}$	$\textbf{0.22} \pm \textbf{0.05}$	5.3±0.2	3.6± 0.4	68

**Table 2.** Chemical and N<sub>2</sub> fixation rate data.

Station numbers correspond to site locations described in Table 1. Samples were collected 15–20 August 2011. Data represent the mean ± the standard error of triplicate samples.



**Fig. 1.** Summer depth-integrated contribution in the Arctic Ocean under low and high open water conditions. The maps depict the time averaged, depth-integrated nitrogen fixed per unit area during the summer months in the Arctic Ocean. The calculation is based on the number of open water days for each location in the Arctic Ocean and the integration is down to a maximum depth of 50 m which includes both the upper mixed layer as well as the seasonal pynocline. The maps depict the amount of nitrogen contributed to the Arctic Ocean between June and September under (A) low (1 standard deviation below the mean condition) and (B) high (1 standard deviation above the mean condition) open water scenarios. The MASIE ice extent data used in the calculation is from June to September (2006–2016). A solid black line bisecting the maps delineates the Western Arctic (WA) longitudinal range used in this study. The remaining Eastern Arctic sector is identified as EA. Sites listed in Table 1 are depicted by red circles and sites sampled by Blais et al. (2012) are depicted by yellow circles. These points provide the spatial range of sites with measured N<sub>2</sub>-fixation rates and the data used in the extrapolations presented in Table 3.

For each of the three domains that we described above (western Arctic shelves, pan-Arctic shelves, and whole Arctic Ocean), the total amount of nitrogen fixed each year was calculated by integrating, in space and time, the measured  $N_2$  fixation rates over the water column depth down to a

maximum depth of 50 m for ice-free days from 01 June to 30 September. These restrictions in time and depth are based on the conditions under which the observations were made (Blais et al. 2012; this study) and our assumptions about the depth range over which the measurements may be valid. All

available measurements were collected under ice-free conditions, within the seasonal surface mixed layer. For the purpose of this extrapolation, we have assumed that rates are constant above 50 m depth because the goal was to determine the maximum contribution potential. Admittedly this may be a source of error if the profile of N<sub>2</sub> fixation rates is a function of depth. Recent *nifH* studies indicate that the majority of Arctic marine diazotrophs are heterotrophic (~ 92%; Fernández-Méndez et al. 2016) and thus may not be directly dependent on light for energy. More studies are required to understand the physiological limitations of Arctic pelagic diazotrophs.

To estimate the potential range of uncertainty introduced by variability in seasonal sea ice cover from year to year, we also calculated the standard deviation of the extent of the seasonal open water area from 2006 to 2016. The standard deviation of the sea ice cover map is then added and subtracted from the mean sea ice cover map, then integrated with the measured N<sub>2</sub> fixation rates to obtain estimates of total fixed nitrogen for both "high" and "low" open water years. Maps of the high (mean + standard deviation) and low (mean – standard deviation) open water days for the entire Arctic are shown in Fig. 1.

#### **Results and discussion**

Active N<sub>2</sub> fixation was found in the coastal Chukchi Sea near Barrow, Alaska with average rates of  $7.7 \pm 1.8$  nmol N  $L^{-1}~d^{-1}$  and  $5.3\pm0.2$  nmol N  $L^{-1}~d^{-1}$  at marine and estuarine sites, respectively (Table 2). A comparison of the  $> 3 \mu m$ and whole water (> 0.7  $\mu$ m) size fractions revealed that approximately 43% of marine and 68% estuarine diazotrophs were  $< 3 \ \mu m$  in size (Table 2). The rates measured in the Chukchi Sea are higher than rates  $(0.01 \pm 0.01 \text{ nmol N})$  $L^{-1} d^{-1}$  to  $4.45 \pm 0.23$  nmol N  $L^{-1} d^{-1}$ ) measured in the Canadian Arctic (Blais et al. 2012). The N<sub>2</sub> fixation rates reported here and in Blais et al. (2012), encompass the area between Baffin Bay on the northeastern boundary and the Chukchi Sea on the northwestern boundary. Assuming uniform N<sub>2</sub> fixation rates in the upper 50 m of the water column, representing the surface mixed layer and the seasonal pycnocline, we estimate that N<sub>2</sub> fixation on the western Arctic shelves can provide 0.01-0.89 mmol N m<sup>-2</sup> d<sup>-1</sup>, which would contribute up to 1.2 Tg N yr<sup>-1</sup> (Table 3). However, most Arctic coastal regions fall within the comparatively wide ranges of temperature  $(-1.2^{\circ}C \text{ to } 19.4^{\circ}C)$ , salinity (4.4-32.4), and nutrient concentrations (< 0.03–6.92  $\mu$ mol nitrate  $L^{-1}$  and 0.17–1.25  $\mu$ mol phosphate  $L^{-1}$ ) where N<sub>2</sub> fixation was observed in the western Arctic (Blais et al. 2012; this study), which suggests that like the conditions under which it was found, N<sub>2</sub> fixation may be more widespread in the Arctic Ocean. Indeed recent investigations using *nifH* gene sequencing found a diverse array of diazotrophs throughout

**Table 3.** Contribution of marine  $N_2$  fixation to local and global nitrogen budgets.

	Ice-free shelves western Arctic	Ice-free shelves entire Arctic
Tg N yr <sup>-1</sup>	$\leq 1.2 \pm 0.3$	$\leq$ 3.5 $\pm$ 0.7
% Annual global N <sub>2</sub> fixation	≤0.9	≤2.7
% Arctic denitrification deficit	0.01–9.3	0.02–27.1

Amount of nitrogen contributed by Arctic N<sub>2</sub> fixation and its potential proportional contributions to annual global N<sub>2</sub> fixation (estimated at 130 Tg N yr<sup>-1</sup>; Eugster and Gruber 2012) and the Arctic denitrification deficit (estimated at 13 Tg N yr<sup>-1</sup>; Chang and Devol 2009) based on two scenarios from June to September: (1) N<sub>2</sub> fixation limited to ice-free shelves in the western Arctic Ocean (1.74 million km<sup>2</sup>) and (2) N<sub>2</sub> fixation present on all ice-free shelves in the entire Arctic Ocean (4.91 million km<sup>2</sup>). The Arctic Ocean is defined as bodies of water that are not rivers or lakes north of 64°N with a water column depth of 1 m or greater. Continental shelves are defined as bodies of water that are not rivers or lakes between 1 m and 200 m in depth; the western Arctic shelf is defined in the longitude range between 175°E and 45°W with water between 1 m and 200 m in depth. The total integrated depth was 50 m or less depending on total water column depth. The ranges reported here are based on the range of marine N<sub>2</sub> fixation rates (0.01-17.15 nmol N  $L^{-1} d^{-1}$ ) reported by Blais et al. (2012) and this study.

the central Arctic and Eurasian basin (Fernández-Méndez et al. 2016).

The Arctic Ocean covers 20% of the continental shelf area in the global ocean. If  $N_2$  fixation occurs across all Arctic shelves, the total amount of  $N_2$  fixed would increase to 3.5 Tg N yr<sup>-1</sup> or 2.7% of global  $N_2$  fixation, based on the current estimate of global marine  $N_2$  fixation of 130 Tg N yr<sup>-1</sup> (Table 3; Eugster and Gruber 2012). Taking this analysis to the extreme, if  $N_2$  fixation is widespread across the entire ice-free Arctic, Arctic marine  $N_2$  fixation could contribute as much as 9.2 Tg N yr<sup>-1</sup> or 7.1% of global marine  $N_2$  fixation in the ice-free summer months.

We note that the estimates described here are based on *net*  $N_2$  fixation rates. Numerous studies have shown that some diazotrophs release upwards of 50% of recently fixed nitrogen as ammonium or dissolved organic nitrogen (e.g., Glibert and Bronk 1994; Benavides et al. 2013). Therefore, the amount of nitrogen released into the ice-free Arctic Ocean is likely more than our reported rates imply.

We also estimated the variability introduced by variation in annual sea ice cover during the summer months. For the entire Arctic Ocean, using the upper range of the observed N<sub>2</sub> fixation rates (17.2 nmol N L<sup>-1</sup> d<sup>-1</sup>), the 1 sigma anomaly in total fixed N<sub>2</sub> is 1.7 Tg N yr<sup>-1</sup> or 18% of the estimate based on average summer sea ice cover. For just the Arctic shelf regions, the 1 sigma anomaly is 0.71 Tg N yr<sup>-1</sup> or 21% of the average, and for the western Arctic only shelf regions, the 1 sigma anomaly is 0.26 Tg N yr<sup>-1</sup> or 22% of the average.

The presence of active N<sub>2</sub> fixation in the Arctic is relevant to outstanding questions of whether the nitrogen budget of the Arctic is balanced. Although some estimates suggest that the annual Arctic nitrogen budget is balanced (Torres-Valdés et al. 2013), measured rates of denitrification on the Arctic shelves are high and produce an estimated nitrogen deficit of 13 Tg N yr<sup>-1</sup> (Chang and Devol 2009), with indirect rate calculations generating higher estimates (Mills et al. 2015). This implies that there is a currently unquantified source of nitrogen to the Arctic Ocean. Although some studies have concluded that N2 fixation is too small to offset any appreciable portion of denitrification (Blais et al. 2012; Mills et al. 2015), our data suggests otherwise. Comparing the estimated denitrification deficit with our N2 fixation estimates described above, N2 fixation within the ice-free western Arctic would offset up to 9.3% of the estimated 13 Tg N yr<sup>-1</sup> Arctic denitrification deficit (Chang and Devol 2009). This increases to 27.1% if N<sub>2</sub> fixation is occurring across all icefree Arctic shelves. Again, taking the extrapolation to the extreme, 70.7% of the Arctic denitrification deficit could be offset by nitrogen fixation if it is active across the entire seasonally ice-free Arctic Ocean (Table 3).

If we consider just the Chukchi shelf, N2 fixation could offset 40% of the denitrification based on direct measurements of denitrification (average measured rate of 0.96 mmol N m<sup>-2</sup> d<sup>-1</sup>; Devol et al. 1997; Chang and Devol 2009) with the average N<sub>2</sub> fixation rates observed at our marine sites  $(0.39 \pm 0.09 \text{ mmol N m}^{-2} \text{ d}^{-1})$ . Although we constrain the depth used for calculations to 50 m based on current measurements (Blais et al. 2012; this study), which are limited to the surface mixed layer, N2 fixation has now been found at sub-euphotic depths (Hamersley et al. 2011; Jayakumar et al. 2012; Rahav et al. 2013). Sub-euphotic N<sub>2</sub> fixation would greatly increase the potential contribution of N2 fixation to the Arctic as would sea-ice and under-ice N<sub>2</sub> fixation (Fernández-Méndez et al. 2016). We have also not included N<sub>2</sub> fixation in Arctic marine sediments in our estimates, which would add a small  $(0.001-0.02 \text{ mmol N m}^{-2} \text{ d}^{-1})$ additional source of fixed nitrogen (Knowles and Wishart 1977; Haines et al. 1981; Gihring et al. 2010). These additional sources would further increase the Arctic's potential as an important source of newly fixed nitrogen and underscore the need for more comprehensive studies investigating the rate of N<sub>2</sub> fixation throughout the Arctic Ocean over broader spatial and temporal scales.

From a global change perspective, the most pressing question is whether  $N_2$  fixation has always occurred in the Arctic or whether it is an emerging phenomenon. The Chukchi Sea has warmed by 0.6°C and the ice-free period has lengthened by 11 d over the last decade (Stroeve et al. 2014). Earlier assumptions that  $N_2$  fixation was restricted to warm and nutrient impoverished euphotic waters have limited watercolumn  $N_2$  fixation studies in high-latitude marine systems. However, more than 50 yr ago  $N_2$  fixation rate measurements were made in the lakes, ponds, and coastal ocean around Barrow, Alaska. Though active N2 fixation was observed in inland waters, no N2 fixation was detected at any of the estuarine or marine sites (Toetz 1961), including the locations where we found active N<sub>2</sub> fixation in this study. The Toetz (1961) study followed <sup>15</sup>N tracer methods described by Dugdale et al. (1959), which differed from the method used in our study. Briefly, the Toetz study aerated samples with an 80 : 20 helium : oxygen mixture under a reduced pressure of 0.8 atm. <sup>15</sup>N<sub>2</sub> gas (95% enriched) was then added to bring the pressure within the sample back up to 1 atm. Any sample with an atom % enrichment higher than 0.386 indicated significant N<sub>2</sub> fixation. With such limited data and a difference in methodological approaches, we cannot say definitively that N<sub>2</sub> fixation is new to the Arctic. However, long-term increases in N2 fixation have been correlated to changes in climate (Sherwood et al. 2014), and the trend towards greater nitrogen limitation in the region makes N<sub>2</sub> fixation more likely.

If N<sub>2</sub> fixation is increasing in the Arctic, cascading changes in the current nitrogen, carbon, and phosphorus cycles may be significant. Current and projected estimates of carbon sequestration in the Arctic are based on the observations that nitrogen, typically nitrate, limits primary production in this region (Tremblay and Gagnon 2009; Popova et al. 2012). As a source of new nitrogen, N<sub>2</sub> fixation could enhance sequestration of carbon dioxide and should be considered in future estimates. Additionally, the Arctic Ocean is a source of phosphorus to the North Atlantic where it supports primary production and N<sub>2</sub> fixation (Yamamoto-Kawai et al. 2006; Torres-Valdés et al. 2013). If Arctic N<sub>2</sub> fixation is providing an additional source of nitrogen, diazotrophy, and the new biomass that it supports could draw down dissolved phosphorus within the Arctic and ultimately reduce the amount of phosphorus reaching the North Atlantic, potentially altering productivity within that region. Increases in Arctic N<sub>2</sub> fixation could thus profoundly affect areal patterns of productivity within both of these oceans. Decreases in soluble phosphorus, in the absence of nitrate, have been observed in regions of the Canadian Arctic (Simpson et al. 2008; Tremblay et al. 2008), providing evidence that this may already be occurring. Clearly, more studies of N2 fixation in Arctic waters are needed. The extrapolations provided in this study highlight the potential contributions of Arctic N<sub>2</sub> fixation and the need to better define its current contribution and to determine how this may change in the future.

#### References

- Arrigo, K. R., and others. 2012. Massive phytoplankton blooms under Arctic sea ice. Science 336: 1408. doi: 10.1126/science.1215065
- Baer, S. E., R. E. Sipler, Q. N. Roberts, P. L. Yager, M. E. Frischer, and D. A. Bronk. 2017. Seasonal nitrogen uptake

and regeneration in the western coastal Arctic. Limnol. Oceanogr. doi:10.1002/lno.10580

- Benavides, M., D. A. Bronk, N. S. Agawin, M. D. Pérez-Hernández, A. Hernández-Guerra, and J. Arístegui. 2013. Longitudinal variability of size-fractionated N2 fixation and DON release rates along 24.5° N in the subtropical North Atlantic. J. Geophys. Res. Oceans **118**: 3406–3415. doi:10.1002/jgrc.20253
- Blais, M., J. É. Tremblay, A. D. Jungblut, J. Gagnon, J. Martin, M. Thaler, and C. Lovejoy. 2012. Nitrogen fixation and identification of potential diazotrophs in the Canadian Arctic. Global Biogeochem. Cycles 26: GB3022. doi:10.1029/2011GB004096
- Bowman, J. S., C. T. Berthiaume, E. V. Armbrust, and J. W. Deming. 2014. The genetic potential for key biogeochemical processes in Arctic frost flowers and young sea ice revealed by metagenomic analysis. FEMS Microbiol. Ecol. 89: 376–387. doi:10.1111/1574-6941.12331
- Chang, B. X., and A. H. Devol. 2009. Seasonal and spatial patterns of sedimentary denitrification rates in the Chukchi Sea. Deep-Sea Res. Part II Top. Stud. Oceanogr. **56**: 1339–1350. doi:10.1016/j.dsr2.2008.10.024
- Dabundo, R., M. F. Lehmann, L. Treibergs, C. R. Tobias, M. A. Altabet, P. H. Moisander, and J. Granger. 2014. The contamination of commercial 15N2 gas stocks with 15N– labeled nitrate and ammonium and consequences for nitrogen fixation measurements. PLoS One 9: e110335. doi:10.1371/journal.pone.0110335
- Devol, A. H., L. A. Codispoti, and J. P. Christensen. 1997. Summer and winter denitrification rates in western Arctic shelf sediments. Cont. Shelf Res. 17: 1029–1050. doi: 10.1016/S0278-4343(97)00003-4
- Dickson, L. G. 2000. Constraints to nitrogen fixation by cryptogamic crusts in a polar desert ecosystem, Devon Island, NWT, Canada. Arct. Antarct. Alp. Res. **32**: 40–45. doi:10.2307/1552408
- Díez, B., B. Bergman, C. Pedrós-Alió, M. Antó, and P. Snoeijs. 2012. High cyanobacterial nifH gene diversity in Arctic seawater and sea ice brine. Environ. Microbiol. Rep. 4: 360–366. doi:10.1111/j.1758-2229.2012.00343.x
- Dugdale, R., V. Dugdale, J. Neess, and J. Goering. 1959. Nitrogen fixation in lakes. Science **130**: 859–860. doi: 10.1126/science.130.3379.859
- Eugster, O., and N. Gruber. 2012. A probabilistic estimate of global marine N-fixation and denitrification. Global Biogeochem. Cycles 26: GB4013. doi:10.1029/2012GB004300
- Fernández-Méndez, M., K. A. Turk-Kubo, P. L. Buttigieg, J. Z. Rapp, T. Krumpen, J. P. Zehr, and A. Boetius. 2016. Diazotroph diversity in the sea ice, melt ponds, and surface waters of the Eurasian Basin of the central Arctic Ocean. Front. Microbiol. 7: 1884. doi:10.3389/fmicb.2016.01884
- Fetterer, F., M. Savoie, S. Helfrich, and P. Clemente-Colón. 2010. Multisensor analyzed sea ice extent-northern hemisphere (MAISE-NH). National Snow Ice Data Center.

- Gihring, T. M., G. Lavik, M. M. Kuypers, and J. E. Kostka. 2010. Direct determination of nitrogen cycling rates and pathways in Arctic fjord sediments (Svalbard, Norway). Limnol. Oceanogr. 55: 740–752. doi:10.4319/ lo.2009.55.2.0740
- Glibert, P. M., and D. A. Bronk. 1994. Release of dissolved organic nitrogen by marine diazotrophic cyanobacteria, Trichodesmium Spp. Appl. Environ. Microbiol. **60**: 3996–4000.
- Großkopf, T., and others. 2012. Doubling of marine dinitrogen-fixation rates based on direct measurements. Nature 488: 361–364. doi:10.1038/nature11338
- Haines, J. R., R. M. Atlas, R. P. Griffiths, and R. Y. Morita. 1981. Denitrification and nitrogen fixation in Alaskan continental shelf sediments. Appl. Environ. Microbiol. 41: 412–421.
- Hamersley, M. R., K. A. Turk, A. Leinweber, N. Gruber, J. P. Zehr, T. Gunderson, and D. G. Capone. 2011. Nitrogen fixation within the water column associated with two hypoxic basins in the Southern California Bight. Aquat. Microb. Ecol. 63: 193. doi:10.3354/ame01494
- Jakobsson, M., and others. 2012. The international bathymetric chart of the Arctic Ocean (IBCAO) version 3.0. Geophys. Res. Lett. **39**: L12609. doi:10.1029/2012GL052219
- Jayakumar, A., M. M. Al-Rshaidat, B. B. Ward, and M. R. Mulholland. 2012. Diversity, distribution, and expression of diazotroph nifH genes in oxygen-deficient waters of the Arabian Sea. FEMS Microbiol. Ecol. 82: 597–606. doi: 10.1111/j.1574-6941.2012.01430.x
- Knowles, R., and C. Wishart. 1977. Nitrogen fixation in Arctic marine sediments: Effect of oil and hydrocarbon fractions. Environ. Pollut. 13: 133–149. doi:10.1016/0013-9327(77)90098-2
- Koroleff, F. 1983. Determination of nutrients, p. 125–187. In K. Grasshoff, M. Eberhardt, and K. Kremling [eds.], Methods of seawater analysis. Verlag Chemie.
- Mills, M. M., and others. 2015. Impacts of low phytoplankton NO  $_3^-$ : PO $_4^{3-}$  utilization ratios over the Chukchi Shelf, Arctic Ocean. Deep-Sea Res. Part II Top. Stud. Oceanogr. **188**: 105–121.
- Mohr, W., T. Grosskopf, D. W. Wallace, and J. Laroche. 2010. Methodological underestimation of oceanic nitrogen fixation rates. PLoS One 5: e12583. doi:10.1371/ journal.pone.0012583
- Moisander, P. H., R. A. Beinart, I. Hewson, A. E. White, K. S. Johnson, C. A. Carlson, J. P. Montoya, and J. P. Zehr. 2010. Unicellular cyanobacterial distributions broaden the oceanic  $N_2$  fixation domain. Science **327**: 1512–1514. doi: 10.1126/science.1185468
- Montoya, J. P., M. Voss, P. Kahler, and D. G. Capone. 1996. A simple, high-precision, high-sensitivity tracer assay for N (inf2) fixation. Appl. Environ. Microbiol. **62**: 986–993.
- Mulholland, M. R., and others. 2012. Rates of dinitrogen fixation and the abundance of diazotrophs in North

Sipler et al.

American coastal waters between Cape Hatteras and Georges Bank. Limnol. Oceanogr. **57**: 1067–1083. doi: 10.4319/lo.2012.57.4.1067

- Parsons, T. R., Y. Maita, and C. Lalli [eds.]. 1984. A manual of chemical and biological methods for seawater analysis. Pergamon Press.
- Popova, E. E., and others. 2012. What controls primary production in the Arctic Ocean? Results from an intercomparison of five general circulation models with biogeochemistry. J. Geophys. Res. Oceans **117**: C00D12. doi:10.1029/2011JC007112
- Rahav, E., E. Bar-Zeev, S. Ohayon, H. Elifantz, N. Belkin, B. Herut, M. R. Mulholland, and I. Berman-Frank. 2013. Dinitrogen fixation in aphotic oxygenated marine environments. Front. Microbiol. 4: 227. doi:10.3389/ fmicb.2013.00227
- Sherwood, O. A., T. P. Guilderson, F. C. Batista, J. T. Schiff, and M. D. Mccarthy. 2014. Increasing subtropical North Pacific Ocean nitrogen fixation since the Little Ice Age. Nature 505: 78–81. doi:10.1038/nature12784
- Shtarkman, Y. M., Z. A. Koçer, R. Edgar, R. S. Veerapaneni, T. D'Elia, P. F. Morris, and S. O. Rogers. 2013. Subglacial Lake Vostok (Antarctica) accretion ice contains a diverse set of sequences from aquatic, marine and sedimentinhabiting bacteria and eukarya. PLoS One 8: e67221. doi: 10.1371/journal.pone.0067221
- Simpson, K. G., J. É. Tremblay, Y. Gratton, and N. M. Price. 2008. An annual study of inorganic and organic nitrogen and phosphorus and silicic acid in the southeastern Beaufort Sea. J. Geophys. Res. Oceans **113**: C07016. doi: 10.1029/2007JC004462
- Sohm, J. A., E. A. Webb, and D. G. Capone. 2011. Emerging patterns of marine nitrogen fixation. Nat. Rev. Microbiol. 9: 499–508. doi:10.1038/nrmicro2594
- Solheim, B., A. Endal, and H. Vigstad. 1996. Nitrogen fixation in Arctic vegetation and soils from Svalbard, Norway. Polar Biol. 16: 35–40. doi:10.1007/BF02388733
- Stroeve, J., T. Markus, L. Boisvert, J. Miller, and A. Barrett. 2014. Changes in Arctic melt season and implications for sea ice loss. Geophys. Res. Lett. **41**: 1216–1225. doi: 10.1002/2013GL058951

- Telling, J., A. M. Anesio, M. Tranter, T. Irvine-Fynn, A. Hodson, C. Butler, and J. Wadham. 2011. Nitrogen fixation on Arctic glaciers, Svalbard. J. Geophys. Res. Biogeosci. 116: G03039. doi:10.1029/2010JG001632
- Toetz, D. W. 1961. Nitrogen fixation in natural waters in the Alaskan Arctic. Masters thesis. Univ. of Wisconsin.
- Torres-Valdés, S., and others. 2013. Export of nutrients from the Arctic Ocean. J. Geophys. Res. Oceans **118**: 1625– 1644. doi:10.1002/jgrc.20063
- Tremblay, J. É., K. Simpson, J. Martin, L. Miller, Y. Gratton, D. Barber, and N. M. Price. 2008. Vertical stability and the annual dynamics of nutrients and chlorophyll fluorescence in the coastal, southeast Beaufort Sea. J. Geophys. Res. Oceans 113: C07S90. doi:10.1029/2007JC004547
- Tremblay, J.-E., and J. Gagnon. 2009. The effects of irradiance and nutrient supply on the productivity of Arctic waters: A perspective on climate change, p. 73–93. *In* J. C. J. Nihoul, and A. G. Kostianoy [eds.], Influence of climate change on the changing Arctic and sub-Arctic conditions. Springer, Dordrecht, Netherlands, doi:10.1007/978-1-4020-9460-6\_7.
- Yamamoto-Kawai, M., E. Carmack, and F. Mclaughlin. 2006. Nitrogen balance and Arctic throughflow. Nature 443: 43–43. doi:10.1038/443043a
- Yu, X., J. Lei, X. Yao, J. Zhu, and X. Jin. 2014. The composition and origination of particles from surface water in the Chukchi Sea, Arctic Ocean. Adv. Polar Sci. 25: 147–154. doi:10.13679/j.advps.2014.3.00147

#### Acknowledgments

We thank L. Killberg-Thoreson and T. Connelly for lab and field assistance, and CH2MHill, UMIAQ science support staff and Barrow Whaling Captains Association for facilitating field work. This work is supported by NSF grant ARC-0909839 to DAB and ARC-1504307 to RES. This paper is Contribution No. 3648 of the Virginia Institute of Marine Science, College of William & Mary.

> Submitted 23 May 2016 Revised 31 January 2017; 08 June 2017 Accepted 10 July 2017